Author's response to reviews

Title: Possible Roles of Epstein-Barr Virus In Castleman Disease

Authors:

Chih-Hao Chen (musclenet2003@yahoo.com.tw)
Hung-Chang Liu (cs520915@yahoo.com.tw)
Tzu-Ti Hung (tzdi0921731610@yahoo.com.tw)
Tsang-Pai Liu (muscle.muscle@msa.hinet.net)

Version: 2 Date: 22 May 2009

Author's response to reviews:

Dear Editor,

Minor changes in the table were made due to inaccuracies. The Manuscript was reviewed by a doctor, a native English speaking person. The questions the reviewer asked were answered. Thanks.